Author's response to reviews

Title: GALA: an international multicentre randomised trial comparing general anaesthesia versus local anaesthesia for carotid surgery

Authors:

Michael J Gough (michael.gough@leedsth.nhs.uk)
Andrew Bodenham (andy.bodenham@leedsth.nhs.uk)
Michael Horrocks (profhorrocks@btclick.com)
Bridget Colam (bridget.colam@ed.ac.uk)
Steff C Lewis (steff.lewis@ed.ac.uk)
Peter Rothwell (peter.rothwell@clinical-neurology.oxford.ac.uk)
Adrian P Banning (Adrian.Banning@orh.nhs.uk)
David Torgerson (djt6@york.ac.uk)
Moira Gough (Moira.Gough@leedsth.nhs.uk)
Demosthenes Dellagrammaticas (mrdemos@ntlworld.com)
Anne Leigh-Brown (Anne.Leigh-Brown@isd.csa.scot.nhs.uk)
Christos Liapis (liapis@med.uoa.gr)
Charles Warlow (charles.warlow@ed.ac.uk)

Version: 2 Date: 8 May 2008

Author's response to reviews:

Affiliation 8 now removed (author included in affiliation 1), and numbers 9 and 10 changed to 8 and 9.

Table heading - legend text placed at base of table

Additional files - information on contents now included at the end of the manuscript.